Author’s response to reviews

Title: Evaluation of preoperative speed of progression and its association with surgical outcomes in primary congenital glaucoma patients: a retrospective study

Authors:

Chunyu Guo (guochunyu2014@gmail.com)
Yue Wu (187297379@qq.com)
Li Xu (x_jenny_dr@hotmail.com)
Mao Li (13995078@qq.com)
Zi Wang (wangzi_827@163.com)
Ni Ni (nini0359@163.com)
Wenyi Guo (wyguo@163.com)

Version: 2  Date: 05 Sep 2017

Author’s response to reviews:

Dear editor,

My colleagues and I would like to submit our revised manuscript entitled Evaluation of preoperative speed of progression and its association with surgical outcomes in primary congenital glaucoma patients: a retrospective study to BMC Ophthalmology.

We appreciate your direction and effort in refining this manuscript, which has been revised per your request.

We hope you find this revised manuscript improved and worthy of publication in BMC Ophthalmology. We are looking forward to your response.
Best regards,

Wenyi Guo, M.D., Ph.D.

Chief physician,

Department of Ophthalmology,

Shanghai Ninth People’s Hospital,

Shanghai JiaoTong University School of Medicine,

639 Zhizaoju Road, Shanghai, 200011, P.R. China

Tel: +86 2163135606

Fax: +86 2163137148

Email: wyguo@163.com